Author's response to reviews

Title: The prevalence of work-related stress, and its association with self-perceived health and sick-leave, in a population of employed Swedish women.

Authors:

      Kristina Holmgren (kristina.holmgren@neuro.gu.se)
    Synneve Dahlin Ivanoff (synneve.dahlin-ivanoff@neuro.gu.se)
  Cecilia Bjorkelund (cecilia.bjorkelund@allmed.gu.se)
Gunnel Hensing (gunnel.hensing@socmed.gu.se)

Version: 5 Date: 11 February 2009

Author's response to reviews: see over
Februari 2009

Dear Philip Dooner,
We have made the listed changes to the format of our manuscript. The title is identical in the manuscript and the submission system, the table order is now correctly labelled and we have removed table and figure placement text.

Best regards,

Kristina Holmgren, PhD
Department of Public Health and Community Medicine
Section of Social Medicine
University of Gothenburg
Box 453
405 30 Gothenburg
Tel +4631-786 68 63
Email kristina.holmgren@neuro.gu.se